Author’s response to reviews

Title: Parents knowledge, attitude and behaviour concerning sunning their babies; a cross-sectional, descriptive study

Authors:

Nihal Aladag (nihal_aladag@yahoo.com)
Muge Filiz (dalim_99@hotmail.com)
Pinar Topsever (topsever@gmail.com)
Suleyman Gorpelioglu (sgorpelioglu@gmail.com)

Version: 3 Date: 16 October 2006

Author’s response to reviews: see over
TO THE EDITOR AND REVIEWERS

The authors thank the reviewers for their contributions. The authors made the (mostly orthographic-grammatical) changes listed below according to the reviewer's (Simone Harrison) suggestion;

1. The word "advice" at pages 2/18 last line, 3/18 second paragraph third line, 4/18 third paragraph 6th line and 8/18 second paragraph 5th line, was changed to "advise".
2. The word "rush" at page 8/18, third paragraph last line was changed to "rash".
3. "have belief" at page 9/18 conclusion section, first line was changed to "believed".
4. Only the valid percentages were presented at Tables 1, 2 and 3, and the number of non respondents were displayed in the footnote section.